Author’s response to reviews

Title: Psychometric properties of the Swedish Childbirth Self-Efficacy Inventory (Swe-CBSEI)

Authors:

Ing-Marie Carlsson (ing-marie.carlsson@hh.se)
Kristina Ziegert (kristina.ziegert@hh.se)
Eva Nissen (eva.nissen@ki.se)

Version: 2 Date: 26 August 2013

Author’s response to reviews:

Authors’ information
The first author, I-MC, is a midwife at Halmstad County Hospital, and doctoral student in reproductive health, at the department of Women’s and Children’s Health at Karolinska University, Sweden and at Halmstad University. The second author, KZ is a registered nurse and associated professor, with research focus in health innovation and supportive processes in health, at School of Social and Health Sciences Halmstad University and at Örebro University. The third author EN, is a midwife and professor at the department of Women’s and Children’s Health at Karolinska University, Sweden. EN is also Core leader and Research manager for Parent-Infant and Breastfeeding Group (PIIB) at karolinska University.